\begin{table}%T2
\caption{\label{tab:Ic_parameters}The parameters of type~Ic SNe.}
\small%\centerline
{
\begin{tabular}{llllllllllll}
\hline\hline\noalign{\smallskip}
SN (Ic)  	& $^{56}$Ni mass 			& Ejecta mass 	& $E_{\rm kin}$ 		& $m_{B,\rm max}$		& $M_{B,\rm max}$$^{a}$	 	&Rise			&$E(B-V)^{b}$	&$\mu$	&Redshift$^{c}$	&Ref.	\\
		&($M_{\odot}$)			&($M_{\odot}$)	&(10$^{51}$~erg)	&				&				&times (d)			&			&		&				&	\\
\hline
2007gr	&  0.076 $\pm$  0.020		&  2.0--3.5		&  1--4			&13.47		&--16.75				&11.5 $\pm$ 2.5	&0.092		&29.84	&0.001728		&1, 2	\\
2004aw   	&  0.30 $\pm$  0.05		&  3.5--8.0		&  3.5--9.0			&18.06		&--17.63				&$\ldots$		&0.370		&34.17	&0.015914		&3	\\
2003jd     	&  0.36 $\pm$ 0.04		&  2.5--3.5	 	&  5--10			&15.75		&--19.30				&$\sim$13		&0.144		&34.46	&0.018860		&4	\\
2002ap	&  0.09, 0.07			&  2.5--5.0	 	&  4--10			&13.11		&--16.76				&$\sim$8		&0.090		&29.50	&0.002192		&5, 6, 7, 8, 9	\\
1998bw	&  0.7, 0.4, 0.5			&  10.9		&  20--50			&14.09		&--18.44				&$\sim$15		&0.060		&32.28	&0.008670		&10, 11, 12, 13  \\
1997ef	&  0.13			&  9.6			&  17.5			&17.45       	&--16.18				&$\ldots$		&0.000		& 33.63	&0.011688		&14	\\
1994I		&  0.07				&  0.88		&  1				&13.77		&--17.06				&$\sim$12		&0.300		&29.60	&0.001544		&15, 16, 17	\\
\hline
\end{tabular}}
\smallskip 
References:  $^1$  \cite{Valenti08a};  $^2$  this paper; $^3$  \cite{Taubenberger06}; $^4$  \cite{Valenti08b}; $^5$ \cite{Yoshii03}; $^6$ \cite{Foley03}; $^7$~\cite{Gal-Yam02}; $^8$ \cite{Tomita06}; $^9$ \cite{Mazzali02}; $^{10}$ \cite{Iwamoto98}; $^{11}$ \cite{Galama98}; $^{12}$ \cite{Nakamura01}; $^{13}$ \cite{Patat01}; $^{14}$ \cite{Mazzali00}; $^{15}$ \cite{Nomoto94}; $^{16}$ \cite{Richmond96}; $^{17}$ \cite{Sauer06}.  
$^{a}$ The peak absolute magnitudes were estimated using 
the apparent magnitudes at maxima. 
  $^{b}$ Total extinction.   $^{c}$ Redshift of the host galaxy (taken from the Nasa/IPAC Extragalactic Database \textit{NED}). 
\end{table}